Author’s response to reviews

Title: Machine Learning Classification of First-Episode Schizophrenia Spectrum Disorders and Controls using Whole Brain White Matter Fractional Anisotropy

Authors:
Pavol Mikolas (pavol.mikolas@gmail.com)
Jaroslav Hlinka (jaroslav.hlinka@nudz.cz)
Antonin Skoch (ansk@medicon.cz)
Pitra Zbynek (z.pitra@gmail.com)
Thomas Frodl (thomas.frodl@med.ovgu.de)
Filip Spaniel (filip.spaniel@nudz.cz)
Tomas Hajek (tomas.hajek@dal.ca)

Version: 2 Date: 06 Feb 2018

Author’s response to reviews:

Dear Dr. Hickey,

thank you for your suggestions. We changed the manuscript accordingly:

1. The corrected passages are now in the section 2.4 Machine learning analyses lines 19-27 and 35-40; section 2.6 Discriminating maps (SVM weight vector) lines 12-19.

2. Section Introduction renamed as Background.

3. Conclusions section added (formerly the last paragraph)
4. Names added and corrected to abbreviations.

5. Figure legend moved after the References and a title added.

6. Cover letter removed.

7. A clean copy uploaded.

Best regards

Pavol Mikolas